Author's response to reviews

Title: A randomised controlled trial of caseload midwifery care: M@NGO (Midwives @ New Group practice Options).

Authors:

Sally K Tracy (sally.tracy@sydney.edu.au)
Donna Hartz (donna.hartz@sydney.edu.au)
Bev Hall (beverley.hall@sydney.edu.au)
Jyai Allen (jyai.allen@mater.org.au)
Amanda Forti (amanda.forti@mater.org.au)
Anne Lainchbury (anne.lainchbury@sesiahs.health.nsw.gov.au)
Jan White (jan.white@sesiahs.health.nsw.gov.au)
Alec Welsh (alec.welsh@unsw.edu.au)
Mark Tracy (mark.tracy@swahs.health.nsw.gov.au)
Sue Kildea (sue.kildea@acu.edu.au)

Version: 5 Date: 21 October 2011

Author's response to reviews: see over
22nd October 2011

The Editor
BMC Pregnancy and Childbirth

Please find our paper attached - a Protocol – outlining the methods and conduct of a randomised controlled trial of caseload midwifery care in Australia.

We have completed the following instructions before resubmitting

Affiliations: Each affiliation written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution

Abstract: reformatted to use these headings: Background, Methods/Design, Discussion.

Structure: the manuscript now has the correct structure for a protocol: Abstract; Background; Methods/Design; Discussion; Abbreviations (if any); Competing interests; Authors' contributions; Authors¿ information (optional); Acknowledgements (optional); References

References: Reference 27 is now cited correctly within the main text and references continue to be cited in numerical order.

Titles: Removed the repetition of the title on page 2 and 3
Thank you again for your consideration
With thanks,

Sally Tracy

Sally K Tracy DMid MA BNURS Adv Dip Mid RGON RM